Author's response to reviews

Title: Increased expression of ACTH (MC2R) and Androgen (AR) receptors in giant bilateral myelolipomas from patients with congenital adrenal hyperplasia

Authors:

Madson Q Almeida (madsonalmeida@usp.br)
Laura Kaupert (laukau@ig.com.br)
Luciana Brito (lucpbrito@gmail.com)
Antonio Lerario (amlerario@gmail.com)
Beatriz Mariani (beatrizmariani@usp.br)
Marta Ribeiro (martinha_ribeiro@hotmail.com)
Osmar Monte (osmarmonte@uol.com.br)
Francisco T Denes (ftdenes@terra.com.br)
Berenice B Mendonca (beremen@usp.br)
Tania Bachega (tbachega@usp.br)

Version: 5  Date: 7 May 2014

Author's response to reviews: see over
Dear Editor-in-Chief:

We have stated the requested information about the written informed consent.

1 - We note that in your manuscript you have included individual patient data and images to accompany this information. In such a case written informed consent to participate in the study is not enough, you would also need to state that written informed consent was obtained for the publication of data and/or images.

Response: We added the following sentence: “informed written consent was obtained from all patients for participate in the study and for the publication of data and/or images”(page 5, line 12).

Sincerely,

Madson Queiroz Almeida, MD
Unidade de Suprarrenal e Endocrinologia do Desenvolvimento, Laboratório de Hormônios e Genética Molecular LIM-42, Hospital das Clínicas e Instituto do Câncer (ICESP), Faculdade de Medicina da Universidade de São Paulo
Av. Dr. Enéas de Carvalho Aguiar, 155, 20 andar Bloco 6
05403-900, São Paulo, SP, Brasil; Phone: 55 11 30697512; Fax: 55 11 30697519
E-mail: madsonalmeida@usp.br